Wed Cct 1 16:54:33 GMI 1997
Year: 97 Doy: 274
Observer: koon
WEATHER COMMENT: Wed Cct 1 16:59: 03 GMI 1997
Cl ear sky, wind=10 nph fromthe SE, tenp=43 F
COMMVENT: Wed Cct 1 17:00: 38 GMI 1997
| had to turn on a few circuit breakers and plug in a power cord to
the MK3 OCP before | could get everything working Iike normal. This
was | eftover fromthe electrical work that was done yesterday by
John Perreira and Steve Sadl er.
Wed Cct 1 17:05:47 GVIT 1997 Pl CS Start Patro
Wed Cct 1 18:00: 15 GVII 1997 Pl CS Fl at
WEATHER COMMENT: Wed Cct 1 18:00:58 GMI 1997
Crrus bl owi ng over.
Wed Cct 1 18:01:16 GMVI 1997 Pl CS End FI at
Wed Cct 1 19:00:17 GVIr 1997 Pl CS Fl at
Wed Cct 1 19:01:15 GMI 1997 PI CS End FIl at
WEATHER COMMENT: Wed Cct 1 19:03:29 GMI 1997
Still have a very thin, continuous |ayer of cirrus or haze, | can see
a rai nbow around the sun.
COMMENT: Wed Cct 1 19:04:42 GV 1997
Adj usted the spar guiding a little so that the MK3 OL gui ding woul d
i mprove
MKI Il COMMENT: Wed Oct 1 19:34:32 GMr 1997
Did a cal despite the bright sky.

Wed Cct 1 19:34:50 GMI 1997 MKI | Start Patro
PICS COWMMENT: Wed Oct 1 19:35:59 GV 1997
Still getting occasional noticeabl e double-inmagi ng on the disk inages,

nostly before 1730 ut, although | saw one around 1900 al so.

MKI Il COMMENT: Wed Oct 1 19:44:04 GJr 1997

| did a cal earlier, data were going to tape before that but now it appears
that data aren’t going to tape, as judged by watching for tape drive activity
at the end of each scan

MKI '] COMMENT: Wed Oct 1 19:56: 00 GVIT 1997

The Wite tape drive field is green all the tine even though no tape

drive activity is visible.

Wed Cct 1 20:00: 17 GVIT 1997 Pl CS Fl at

Wed COct 1 20:01:16 GVII 1997 Pl CS End FI at

COMMENT: Wed Cct 1 20:43:08 GV 1997

Lost spar guiding when it hit the |adder, fixed it.

Wed Oct 1 20:59:05 GVII 1997 MKI T Start Patro
Wed Cct 1 21:00:16 GVII 1997 Pl CS Fl at
Wed Oct 1 21:01:14 GVII 1997 Pl CS End FIl at

MKI 1| COMMENT: Wed Oct 1 21:05:16 GV 1997
Tried stopping scans with Idle, then restarted but data still not going



tota

PICS COMMENT: Wed Cct

pe. Stopping scans due to increasingly bright sky.

1 21:38:14 GVII 1997

Pausing to work on Spar and Dome stuff.

atro

COMMENT: Wed Cct
doing Scintilloneter data dunp,
COMMENT: Thu Cct
cl eaned,

MKIT |

Rernoved,

Activity report:
ronal activity.

No co

TAPES:

* Kk kkk*k

MKI T |

Pl CS

. HO1651
CH P: Down for filter

P01001

repair.

LOAL: LO0476 in drive #0

Thu Cct

16_57
17_01
17_03
17_06
17_09
17_12
17_15
17_18
17_21
17_24
17_27
17_30
17_33
17_36
17_39
17_42
17_44

. rawnk3
. rawnk3
. rawrk3
. rawrk3
. rawnk3
. rawnk3
. rawnk3
. rawnk3
. rawrk3
. rawrk3
. rawnk3
. rawnk3
. rawnk3
. rawnk3
. rawrk3
. rawrk3
. rawnk3

17_47.
17_50.
17_53.
17_56.
17_509.
18_02.
18_05.
18_08.
18_11.
18_14.
18 17.
18_19.
18~ 22.
18_25.
18_28.
18_31.
18_34.

2 00:38:21 GVIr 1997
M1 T

r awnk3
r awnk3
rawk3
rawrk3
rawrk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawrk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawrk3

18_37.
18_40.
18_43.
18_46.
18~ 49.
18_52.
18_55.
18_57.
19_00.
19_03.
19_06.
19_09.
19_15.
19_22.
19_28.
19_34.
19_37.

will
2 00:36:56 GVIT 1997
and reinstalled O1.

1 21:58:45 GV 1997

mai |

r awnk3
r awnk3
rawk3
rawrk3
rawk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawk3

in Hlo.

19_40.
19_43.
19_46.
19_49.
19_52.
19_55.
19_58.
20 01.
20_04.
20 07.
20_09.
20 12.
20_15.
20_18.
20 _21.
20 24.
20 27.

r awnk3
r awnk3
rawk3
rawrk3
rawrk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawk3
r awrk3
r awnk3
r awnk3
rawk3
rawrk3
rawk3

Cl eaned and strai ghtened up done.

20 _30. rawnk3
20 _33. rawnk3
20_36. rawrk3
20_39. rawrk3
20_42. rawmrk3
20_45. rawmrk3
20 _47. rawk3
20 _50. rawnk3
20_53. rawrk3
20_59. rawrk3
21_02.rawrk3
21_05. rawmrk3
21 07.rawk3
cl1l9 12. rawrk3
c19 18.rawnk3
cl19 25.rawnk3



